Author's response to reviews

Title: HIV serostatus and disclosure: implications for infant feeding practice in rural south Nyanza, Kenya

Authors:

Maricianah A Onono (maricianah@yahoo.com)
Craig R Cohen (ccohen@globalhealth.ucsf.edu)
Mable Jerop (jeropmable@gmail.com)
Elizabeth A Bukusi (ebukusi@kemri.org)
Janet M Turan (jmturan@uab.edu)

Version: 2 Date: 12 June 2013

Author's response to reviews:

In response to your email dated 12 June 2013, please find enclosed the revised manuscript, “HIV serostatus and disclosure: implications for infant feeding practice in rural south Nyanza, Kenya” by Onono Maricianah et al. As requested we have now included the following statement on ethical approval under the methods section:

“The Committee on Human Research at UCSF and the Ethical Review Committee at KEMRI in Kenya approved this study. “ This sentence is the first sentence in the subsection labelled study procedures on page 7.